Author’s response to reviews

Title: Spontaneous Posterior Rectus Sheath Hernia: Case Report

Authors:

Chu Woon Ng (chuwoonn@yahoo.com)

Anna Sandstrom (Anna.Sandstrom@health.qld.gov.au)

Grace Lim (Grace.Lim@health.qld.gov.au)

Version: 2 Date: 05 Mar 2018

Author’s response to reviews:

More information is needed:

The case report should include past medical, social, environmental, family and employment history.

Report updated with the requested information.

What medications was the patient on prior to diagnosis? Did the patient smoke, and/or consume alcohol?

Report updated with the requested information.

Give detailed physical and neurological examination on admission. What was the temperature, pulse, blood pressure and temperature, on admission?

Report updated with the requested information.

Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc

Report updated with the requested information (only the lab results that was not normal).

Give information about follow-up for at least 6 months.

Patient was followed up at outpatient clinic two weeks after he discharged and no further follow up indicated. He is under conservative management at GP level.